Response to the Editor/reviewer’s comment

Thank you for giving us the opportunity to submit a revised draft of the manuscript. We are most grateful for the time the editors and reviewers spent on providing suggestions on how to improve our paper. In our revision, we have tried to address all the queries raised by the editor and the reviewers’.

Reviewers comments:

Reviewer #1: It is an interesting topic. I am pleased to accept this manuscript.

Reviewer #2: no comment. This manuscript has the potential to be accepted.

Response: Thank you for the kind words of appreciation. We acknowledge the great effort of both the editors and the reviewers in giving us important insights and suggestions to help improve our paper’s quality. We are most thankful for the time they invested in going through, assessing and approving our paper. Since there are no major/specific concerns raised by either of the reviewer, I am returning the manuscript unchanged after proof reading the document.

Once again we express our sincere thanks and gratitude to all the editor and the reviewers comment for providing valuable insights into our paper. we believe that these comments indeed go a long way in improving the overall quality and clinical impact of our research in the future.